All Rights Reserved.

© 2021. American Geophysical Union.



Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL091753

Jiamin Wang and Yuping Guan contributed equally to this work.

Key Points:

- Climate change has driven longer and hotter summers, shorter and warmer winters, shorter springs and autumns
- The onsets of spring and summer are advanced, while the onsets of autumn and winter are delayed
- Such changes in four seasons can be mainly attributed to greenhousewarming, and will be amplified under the business-as-usual scenario

Supporting Information:

- Supporting Information S1
- Movie S1

Correspondence to: Y. Guan, guan@scsio.ac.cn

Citation:

Wang, J., Guan, Y., Wu, L., Guan, X., Cai, W., Huang, J., et al. (2021). Changing lengths of the four seasons by global warming. *Geophysical Research Letters*, 48, e2020GL091753. https://doi. org/10.1029/2020GL091753

Received 17 NOV 2020 Accepted 10 FEB 2021

Changing Lengths of the Four Seasons by Global Warming

Jiamin Wang^{1,2}, Yuping Guan², Lixin Wu^{3,4}, Xiaodan Guan¹, Wenju Cai^{3,4,5}, Jianping Huang¹, Wenjie Dong^{6,7}, and Banglin Zhang⁸

¹Key Laboratory for Semi-Arid Climate Change of the Ministry of Education, College of Atmospheric Sciences, Lanzhou University, Lanzhou, China, ²State Key Laboratory of Tropical Oceanography, South China Sea Institute of Oceanology, Chinese Academy of Sciences, Guangzhou, and College of Marine Sciences, University of Chinese Academy Sciences, Beijing, China, ³Physical Oceanography Laboratory/Institute for Advanced Ocean Study, Ocean University of China, Qingdao, China, ⁴Pilot National Laboratory for Marine Science and Technology (Qingdao), Qingdao, China, ⁵Centre for Southern Hemisphere Oceans Research, CSIRO Oceans and Atmosphere, Hobart, TAS, Australia, ⁶School of Atmospheric Sciences, Sun Yat-Sen University, Zhuhai, and Key Laboratory of Tropical Atmosphere-Ocean System, Ministry of Education, Zhuhai, China, ⁷Southern Marine Science and Engineering Guangdong Laboratory (Zhuhai), Zhuhai, China, ⁸Guangzhou Institute of Tropical and Marine Meteorology/ Guangdong Provincial Key Laboratory of Regional Numerical Weather Prediction, CMA, Guangzhou, China

Abstract How long will the four seasons be by 2100? Increasing evidence suggests that the length of a single season or in regional scales has changed under global warming, but a hemispherical-scale response of the four seasons in the past and future remains unknown. We find that summer in the Northern Hemisphere midlatitudes has lengthened, whereas winter has shortened, owing to shifts in their onsets and withdrawals, accompanied by shorter spring and autumn. Such changes in lengths and onsets can be mainly attributed to greenhouse-warming. Even if the current warming rate does not accelerate, changes in seasons will still be exacerbated in the future. Under the business-as-usual scenario, summer is projected to last nearly half a year, but winter less than 2 months by 2100. The changing seasonal clock signifies disturbed agriculture seasons and rhythm of species activities, more frequent heat waves, storms and wildfires, amounting to increased risks to humanity.

Plain Language Summary A series of phenomena such as early flowering of plants and early migratory birds are suggesting that the traditional four seasons may have changed. We focus on how the four seasons changed during 1952–2011 and will change by the end of this century in the warming Northern Hemisphere midlatitudes. We find that lengths and start dates of the four seasons have changed, and the changes will be amplified in the future. Over the period of 1952–2011, the length of summer increased from 78 to 95 days and that of spring, autumn and winter decreased from 124 to 115, 87 to 82, and 76 to 73 days, respectively. In addition, summer is projected to last nearly half a year, but winter less than 2 months by 2100. Such changes can trigger a chain of reactions in agriculture, policy-making for agricultural management and disaster prevention requires adjustment accordingly. The seasonal-related topics involving ecology, the ocean and the atmosphere also need to be revisited.

1. Introduction

As global mean temperature rises, climate-related impacts and risks are increasing rapidly (Hoegh-Guldberg et al., 2019). One prominent example is that global warming has caused the phase and temperature amplitude of seasons to shift significantly, which have been detected in both temperature (Christidis et al., 2007; Menzel & Fabian, 1999; Park et al., 2018; Peña-Ortiz et al., 2015; Santer et al., 2018; Yan et al., 2011) and phenological indicators (Allstadt et al., 2015; Schwartz et al., 2006). The changes in phase of the four seasons include the expansion and contraction of lengths and the drift of onsets (Stine et al., 2009). Therefore, seasonal cycles of the Earth's climate at temperate latitudes are no longer appropriately defined by dividing the year into four equal-length seasons (Trenberth, 1983). Many signals in ecological and physical systems have been subjected to seasonal changes. The most obvious are related to rhythms of living things; for instance, traditional phenological cycles of plants (Buermann et al., 2018; Reyes-Fox et al., 2014; Steltzer & Post, 2009) and bird migration patterns have shifted in response to seasonal changes (Both et al., 2006;

Horton et al., 2020). Temporal and spatial mismatches among organisms are also gradually exaggerating because not all species cope with seasonal changes at the same rate or in the same direction (Samplonius & Both, 2019), which disrupts the structure and function of ecological communities. Even more serious is that seasonal changes can magnify the strain on humans imposed by global warming. People are exposed to more allergenic pollen and longer allergic periods owing to ongoing extension of the growing season (Anenberg et al., 2017). In addition, tropical mosquitoes carrying viruses are likely to expand northward and bring about explosive outbreaks during longer and hotter summers, especially when introduced into regions without previous occurrence (Ryan et al., 2019). These impacts caused by variations in seasons heighten the urgency of understanding how the phase and temperature of seasons change and whether they will continue in the future (Paniw et al., 2019).

While previous studies have quantified observed variations in length of a single season or in regional scales (Christidis et al., 2007; Park et al., 2018; Peña-Ortiz et al., 2015; Yan et al., 2011), a hemispherical-scale response of the four seasons is lacking. Here, we consider how the four seasons changed during 1952–2011 and will change by the end of this century in the warming Northern Hemisphere midlatitudes (30–60°N), where the division of the four seasons based on temperature is readily available. At low latitudes (0–30°N), the air temperatures vary so little throughout the year that the differences in temperature among the four seasons are not obvious (Santer et al., 2018). Tropical seasons are generally divided into wet seasons and dry seasons according to precipitation. Consequently, based on temperature, division of the year into four seasons at low latitudes is not appropriate, so are high latitudes (60–90°N), where the winter is long and cold, and the summer is short and warm (Trenberth, 1983).

2. Data and Methods

2.1. Data

Temperature data in this study were obtained from observations and multimodel ensemble simulations. The observed data were obtained from the Hadley Centre's Global Historical Climatology Network Daily (HadGHCND) and include gridded data of daily maximum and minimum temperatures over land with a horizontal resolution of 3.75° longitude × 2.5° latitude (Caesar et al., 2006). Daily mean observed temperature for the period 1951-2011 was calculated as the average of the maximum and minimum. The multimodel ensemble simulations for the period 1951-2100 were obtained from the Coupled Model Intercomparison Project 5 (CMIP5) and 6 (CMIP6). For CMIP5, two different representative concentration pathways (RCPs) were selected. RCP4.5 is a stabilization scenario, with the total radiative forcing rising until 2070 followed by stable concentrations (without an overshoot pathway) to 4.5 W m^{-2} after 2070 (Thomson et al., 2011). RCP8.5 is a continuously rising radiative forcing pathway in which the radiative forcing level by the end of the 21st century will be approximately 8.5 W m⁻² (Riahi et al., 2011). SSP2-4.5 and SSP5-8.5 were selected for CMIP6. For SSPx-y, SSP represents shared socioeconomic pathways, and x and y represent the specific SSP and the forcing pathway, respectively. Specifically, SSP2-4.5 represents the middle of the future forcing pathway range, and SSP5-8.5 represents the high end of the future forcing pathway range. SSP2-4.5 and SSP5-8.5 update the RCP4.5 and RCP8.5 pathways, respectively (Gidden et al., 2019; O'Neill et al., 2016). Twenty-one global climate models from CMIP5 and 16 from CMIP6 were considered in our study (Figure S1) and include both historical climate simulations (1951–2011) and climate projections (2015–2100). Note that the historical climate simulations (1951-2011) of CMIP5 were constructed by merging historical experiments (1951-2005) with future projections based on the RCP4.5 scenario (2006-2011).

2.2. Definition of the Four Seasons

For convenience of analysis, the temperatures on February 29 of leap years were excluded to restrict each year to 365 days, which had no effect on the calculation of long-term trends. To redivide the four seasons, we defined local temperature thresholds following previous methods (Christidis et al., 2007; Park et al., 2018) as outlined below. Summer started when the temperature exceeded the 75th percentile of temperature averaged over 1952–2011 and ended when the temperature was below the 75th percentile. The temperature threshold for winter was defined as the 25th percentile. Spring was regarded as the transition from winter to summer with increasing temperature, and autumn was defined as the transition from summer to winter





Figure 1. Spatial distributions of the linear trends in season lengths and onsets during the period of 1952–2011. Seasonal distributions of spring (a and e), summer (b and f), autumn (c and g), and winter (d and h). A negative (positive) onset means an advance (delay) in seasonal start. Dotted grids indicate statistically significant trends at the 5% level based on Student's *t*-test.

with decreasing temperature. A third-degree polynomial fitted to the raw observed data was employed to smooth day-to-day temperature fluctuations to avoid more than two intersections between each temperature threshold and observed daily temperature (Christidis et al., 2007; Park et al., 2018) (Figure S2). The temperature thresholds varying from regions to regions well reflect the response of the four seasons to swift climate change and the differences in the four seasons among different regions. Values related to seasonal variations from both observations and model data were computed with the same thresholds.

3. Results

3.1. Observed Changes in Seasonal Cycles

To provide a comprehensive explanation of variations in the four seasons under climate warming, we first calculated the changed linear trends in lengths from 1952 to 2011 in the Northern Hemisphere midlatitudes. For the observed, we found that the hemispheric-scaled summer expanded, spring, autumn and winter contracted (Figures 1a–1c and 1d). The trends of the four seasons in some areas of North America were contrary to the overall trends (Figures 1a–1c and 1d), which is likely attributed to the strong influence of internal climate variability (Huang et al., 2017). In the study area, the most prominent changes in seasonal durations were in the direction of longer summers and shorter winters. From 1952 to 2011, summer length extended at an average rate of 4.2 days every 10 years (d/10 years), while winter length shortened at an average rate of 2.1 d/10 years (Table S1). The durations of spring and autumn changed slightly, shrinking at rates of 1.0 and 1.1 d/10 years, respectively (Table S1). The most obvious area of changes in lengths of the four seasons is near the Mediterranean region, which is consistent with the strengthening seasonality around the Mediterranean region since 1950s in the context that seasonality is weakening in Northern Hemispheric mid-high latitudes (Qian & Zhang, 2019). The Tibet Plateau and its western regions also show greater changes.

As the lengths of the four seasons changed, so did the onsets of the four seasons (Figures 1e–1g and 1h). Spring and summer started earlier by 1.6 days/10 years and 2.5 days/10 years, respectively, while the onset of autumn and winter was delayed by 1.7 days/10 years and 0.5 days/10 years, respectively (Table S2). The changes in onsets showed apparent regional differences. The onsets of spring, summer and autumn in western Eurasia exhibited the strongest trends (Figures 1e–1g), and the delay of the start of winter was the most pronounced on the Qinghai Tibetan Plateau (Figure 1h), which is consistent with the spatial distribution of the four seasons lengths. Therefore, the variations in lengths of the four seasons are the outcomes of the displacement of seasonal onsets. The accurate onsets of the four seasons are more instructive to the cyclical



Figure 2. The Spatial distributions of the linear trends in temperatures within the lengths of four seasons over 1952–2011. Seasonal distributions of spring (a and e), summer (b and f), autumn (c and g), and winter (d and h). Dotted grids indicate statistically significant trends at the 5% level based on Student's *t*-test.

events of animal and plant phenology. For example, breeding birds need select a breeding window to feed their offspring with adequate food and avoid enemies. However, due to the early spring, breeding birds breed early and the breeding window is becoming narrower, which commit them to severe food shortages and threats from predators (Descamps et al., 2019).

Variations in both seasonal phases and seasonal temperatures have been detected under the temperature threshold definition. Temperature has risen clearly in extended summers and shortened winters from 1952 to 2011 (Figures 2b and 2d). In summer, the temperature increased at the rate of 0.089°C per 10 years (°C/10 years) (Table S3). Since the temperature threshold that define each season are certain, the temperature variation in summer is consistent with the fact that extreme high-temperature events have occurred more frequently in summers (Sheridan & Lee, 2018). In the extended summer season, from May to September in the Northern Hemisphere, the frequency, duration and cumulative heat of heatwave have significantly increased since 1950s (Perkins-Kirkpatrick & Lewis, 2020). Hotter and longer summers seriously disrupt energy productivity and utilization (Orlov et al., 2020), such as increasing the energy demand for cooling and affecting hydropower generation. The summers also plague vegetation and forests, higher large-wild-fire frequency and longer wildfire seasons during heatwaves (Westerling et al., 2006). In addition, existing evidence suggests that high temperature is even tied to human physical and mental health. Groups who work outdoors or indoors without cooling systems are facing severe heat exhaustion (Kjellstrom, 2015). The same is true for mental health: the higher the temperature, the higher the prevalence of mental health (Obradovich et al., 2018).

In winter, the temperature increased at the rate of 0.260°C/10 years (Table S3). The temperatures in the whole Eastern Hemisphere have changed by more than 0.2°C/10 years, especially in eastern East Asia. The warming trend of winter was most pronounced in northern North America, where the temperature rose at the rate of more than 0.4°C/10 years (Figure 2d). Warmer winters have a great impact on crop yields. The insufficient chilling demand for bud dormancy due to milder winters stunt the growth of plants, resulting in reduced yields and quality (Bartolini et al., 2019; Fu et al., 2015). In addition, milder and shorter winters come with great challenges visible in public security. Higher temperatures and shorter winter periods cause higher crime rates across several regions of the United States (Harp & Karnauskas, 2018). Tourism also has also faced the pressure from winter changes. Highly snow-dependent ski areas are most easily and directly affected by winter (Beniston, 2003).

However, the temperature in spring decreased at the 95% confidence level (Figure 2b). Increasingly colder and earlier spring seasons may pose a greater risk of false spring (subsequent hard freezes damage plants that awaken prematurely from dormancy) for plants (Allstadt et al., 2015), which may inflict a heavy blow to the economy (Kral-O'Brien et al., 2019). The false spring in Michigan in 2012 resulted in more than half a billion dollars loss from local fruit trees (Knudson, 2012). Plants that survive false springs may not be able to grow successfully due to weak resistance to future stress (Allstadt et al., 2015). The seasonal effect is not limited to acting on itself; it spreads to other seasons and even brings about long-term consequences; for instance, early springs are likely to indicate early summers (Crimmins & Crimmins, 2019), early growth





Figure 3. Temporal trends for onsets and lengths of the four seasons over the 1952–2011 period from observations of HadGHCND and different experiments of CMIP6. Seasonal trends in lengths (a–d) and onsets (e–h) of spring (a and e), summer (b and f), autumn (c and g), and winter (d and h). CMIP6-ALL represents all forcing simulation, CMIP6-GHG represents well-mixed greenhouse gases run, CMIP6-AER represents anthropogenic aerosols forcing run and CMIP6-NAT represents natural forcing run. The uncertainty is calculated as double standard errors, which reflects the 95% confidence interval. HadGHCND, Hadley Centre's Global Historical Climatology Network Daily; CMIP6, Coupled Model Intercomparison Project 6.

due to advanced onset of spring may face a shortage of water in the remainder of the growing season (Buermann et al., 2018), and short and warm winters can exacerbate the occurrence of false springs (Allstadt et al., 2015). The chain reactions complicate the effects of seasonal changes. Robust seasonal forecasts will enable the optimization of the operation and management of activities, such as agricultural production, and avoid some possible risks.

To find the contributions of different factors including greenhouse gases, anthropogenic aerosols, and natural forcing to the observed trends in lengths and onsets of the four seasons, we compared multimodel mean trends of the lengths and onsets of the four seasons with the observations (Table S4 and Figure 3). The trends from all forcing simulation show similar trend to the observations, and the trends forced by only well-mixed greenhouse gases are similar to all forcing simulation. However, the simulations ran by only anthropogenic aerosols forcing or only natural forcing produce weak or even contrary trends. The results suggest that greenhouse gases forcing dominate changes in lengths and onsets of the four seasons. The changes in lengths and onsets of the four seasons are also calculated based on detrended observational data (Figure S3). After removing warming trends, the changes in seasonal cycles are greatly weakened, which also indicate that greenhouse-warming trends induced variations in the four seasons.

3.2. Projected Changes in Seasonal Cycles

Previous studies have not assessed future changes in the four seasons of the midlatitude Northern Hemisphere, as such, whether these changes continue is an important issue. We use the same definition of seasonal cycle as in the observed to project changes in seasonal cycles in the future. We first evaluated CMIP5 and CMIP6 models using the Taylor diagram analysis, and selected 21 CMIP5 models and 16 CMIP6 models whose historical experiments all reproduced the temperature climatology well (Figure S1). Both CMIP5 and CMIP6 ensemble simulations from historical experiments also reproduced the changed seasonal trend direction well (Figure 4, Tables S1–S3, and Figures S4–S6). Based on the multimodel ensemble 2-m surface air temperature from CMIP5 and CMIP6, the changed trends of the four seasons are expected to amplify in the future. Under the stabilization scenario (RCP4.5 and SSP2-4.5), trends in length and onset from 2016 to 2100 are close to the observed extrapolation (the trends in length and onset of the four seasons during 2016–2100 are the same as observed trends from HadGHCND during 1952–2011) (Figure 4 and Figures S7–S8), indicating that even if seasons are to continue at the current rate observed, a longer summer and shorter winter will become the new normal in the 21st century.





Figure 4. Temporal trends for onsets, lengths and temperatures of the four seasons during the period of 1952–2011 (historical) and 2016–2100 (future). (a) Onsets are shown in (a). The shading denotes the range of traditional seasons (green: spring, pink: summer, khaki: autumn, blue: winter). (b) Lengths and temperatures are shown in (b) The curve is the length trend, and shading is the temperature trend from HadGHCND and RCP8.5 (trends for others are listed in Table S3). The labels on the *Y* axis in parentheses indicate coordinates of the graph above, and labels outside the parentheses indicate the coordinates below. HadGHCND, Hadley Centre's Global Historical Climatology Network Daily; RCP, representative concentration pathways.

With radiative forcing levels rising, spring and summer will start earlier, with respective mean rates of 3.3 days/10 years and 4.6 days/10 years under RCP8.5, while autumn and winter will start later, by 3.8 days/10 years and 1.4 days/10 years, consistent with the variations under SSP5-8.5 (Tables S1–S2, Figures S10–S11). The shifts in onsets of the four seasons will cause the length of spring, autumn and winter to shrink at a rate of 1.3, 2.4, 4.7 days/10 years and summer to extend at 8.5 days/10 years (Table S1). Except for spring, the length of the other three seasons under RCP8.5 is more than twice as long as the observed. As a result, there is good potential for a 166-days summer and 31-days winter in 2100 (Figure 5 and Table S5). The most aggressive scenario alarms us that we will be on a dangerous path of agriculture, ecology, and human health for climate change if emissions are not curbed (Christidis et al., 2020; Power & Delage, 2019). People will also face a higher risk of seasonal temperature in the future (Figures S9 and S12), especially in summer and winter, when the temperature trend will be nearly three times that during the period of 1952–2011 under the scenario of RCP8.5 and SSP5-8.5 (Figures 4b and Table S3). The changes may magnify the impact of the four seasons, and even cause more severe consequences. Seasonal change is a global phenomenon; however, the magnitudes are not uniform across spatial scales, and specific regional research is necessary in the future.

To examine sensitivity to definition of seasons, the possible influence of temperature thresholds based on climatology from different periods and defined by different percentiles has been noticed. We have calculated variations in lengths, onsets and temperatures of the four seasons under different definitions whose temperature thresholds defined by temperature averaged over different periods, such as 1952–2011, 1970–1999, 1952–1981, and 1982–2011 (Figures S13–S18 and Tables S6–S8). The results are insensitive to reasonable definitions. Similar results are also observed when temperature thresholds are defined by different percentiles (25th/75th, 25th/65th, 25th/70th, 25th/80th, 20th/75th, 30th/75th, 35th/75th). Therefore, the results are robust to temperature thresholds based on climatology from different periods and defined by different percentiles.



Geophysical Research Letters



Figure 5. Onsets and lengths of the four seasons in 1952, 2011, 2050, and 2100. The top row is from HadGHCND, and the bottom row is from RCP8.5. HadGHCND, Hadley Centre's Global Historical Climatology Network Daily; RCP, representative concentration pathways.

4. Conclusions

As global warming intensifies, the four seasons of a year no longer have equal months, and their onsets are irregular. Over the period of 1952–2011, the length of summer in the Northern Hemisphere midlatitude increased from 78 to 95 days and that of spring, autumn and winter decreased from 124 to 115, 87 to 82 and 76 to 73 days, respectively (Table S5). Accordingly, seasonal temperature also changed, with summer and winter becoming warmer. Longer and hotter summers, shorter and warmer winters, shorter spring and autumn seasons are the new normal, and this kind of trend may be unavoidably amplified in the future due to the rising radiative forcing. Variations of the four seasons simulation from CMIP5 and CMIP6 under different scenarios exhibit a range of plausible future four seasons, providing decision-makers with a broader basis for decision-making. Under the business-as-usual scenario, spring and summer will start about a month earlier than 2011 by the end of the century, autumn and winter start about half a month later, which result in nearly half a year of summer and less than 2 months of winter in 2100. As lengths of



the four seasons change continue, which can trigger a chain of reactions, policy-making for agricultural management, health care, and disaster prevention requires adjustment. Above all, seasonal-related topics involving ecology, the ocean and the atmosphere need to be revisited because seasons are the basic time parameter for a wide range of natural phenomena (Cassou & Cattiaux, 2016).

Data Availability Statement

The HadGHCND data are available for download (https://www.metoffice.gov.uk/hadobs/hadghcnd/ download.html). The CMIP5 and CMIP6 outputs are downloaded from (https://esgf-node.llnl.gov/search/cmip5/; https://esgf-node.llnl.gov/search/cmip6/).

References

- Allstadt, A. J., Vavrus, S. J., Heglund, P. J., Pidgeon, A. M., Thogmartin, W. E., & Radeloff, V. C. (2015). Spring plant phenology and false springs in the conterminous US during the 21st century. *Environmental Research Letters*, *10*, 104008. https://doi.org/10.1088/1748-9326/10/10/104008
- Anenberg, S. C., Weinberger, K. R., Henry, R., James, E. N., Allison, C., Neal, F., et al. (2017). Impacts of oak pollen on allergic asthma in the United States and potential influence of future climate change. *GeoHealth*, 1, 80–92. https://doi.org/10.1002/2017GH000055
- Bartolini, S., Massai, R., Iacona, C., Guerriero, R., & Viti, R. (2019). Forty-year investigations on apricot blooming: Evidences of climate change effects. *Scientia Horticulturae*, 244(2019), 399–405. https://doi.org/10.1016/j.scienta.2018.09.070
- Beniston, M. (2003). Climatic change in mountain regions: A review of possible impacts. *Climate Change*, 59, 5–31. https://doi.org/10.1023/A:1024458411589
- Both, C., Bouwhuis, S., Lessells, C. M., & Visser, M. E. (2006). Climate change and population declines in a long-distance migratory bird. *Nature*, 441, 81–83. https://doi.org/10.1038/nature04539
- Buermann, W., Forkel, M., O'Sullivan1, M., Sitch, S., Friedlingstein, P., & Haverd, V. (2018). Widespread seasonal compensation effects of spring warming on northern plant productivity. *Nature*, 562, 110–115. https://doi.org/10.1038/s41586-018-0555-7
- Caesar, J., Alexander, L., & Vose, R. (2006). Large-scale changes in observed daily maximum and minimum temperatures: creation and analysis of a new gridded data set. Journal of Geophysical Research, 111, D05101. https://doi.org/10.1029/2005JD006280
- Cassou, C., & Cattiaux, J. (2016). Disruption of the European climate seasonal clock in a warming world. *Nature Climate Chang*, 6(6), 589–594. https://doi.org/10.1038/nclimate2969
- Christidis, N., McCarthy, M., & Stott, P. A. (2020). The increasing likelihood of temperatures above 30 to 40°C in the United Kingdom. *Nature Communication*, *11*(1), 3093. https://doi.org/10.1038/s41467-020-16834-0
- Christidis, N., Stott, P. A., Brown, S., Karoly, D. J., & Caesar, J. (2007). Human contribution to the lengthening of the growing season during 1950–99. *Journal of Climate*, 20, 5441–5454. https://doi.org/10.1175/2007JCLI1568.1
- Crimmins, M. A., & Crimmins, T. M. (2019). Does an early spring indicate an early summer? relationships between intraseasonal growing degree day thresholds. *Journal of Geophysical Research: Biogeosciences*, 124, 2628–2641. https://doi.org/10.1029/2019JG005297
- Descamps, S., Ramírez, F., Benjaminsen, S., Anker-Nilssen, T., Barrett, R. T., Burr, Z., et al. (2019). Diverging phenological responses of Arctic seabirds to an earlier spring. *Global Change Biology*, *25*(49), 4081–4091. https://doi.org/10.1111/gcb.14780
- Fu, Y. H., Zhao, H., Piao, S., Peaucelle, M., Peng, S., Zhou, G. M., et al. (2015). Declining global warming effects on the phenology of spring leaf unfolding. *Nature*, 526, 104–107. https://doi.org/10.1038/nature15402
- Gidden, M. J., Riahi, K., Smith, S. J., Fujimori, S., Luderer, G., Kriegler, E., et al. (2019). Global emissions pathways under different socioeconomic scenarios for use in CMIP6: A dataset of harmonized emissions trajectories through the end of the century. *Geoscientific Model Development*, 12, 1443–1475. https://doi.org/10.5194/gmd-2018-266
- Harp, R. D., & Karnauskas, K. B. (2018). The influence of interannual climate variability on regional violent crime rates in the United States. *GeoHealth*, 2(11), 356–369. https://doi.org/10.1029/2018GH000152
- Hoegh-Guldberg, O., Jacob, D., Taylor, M., Guillén Bolaños, T., Bindi, M., Brown, S., et al. (2019). The human imperative of stabilizing global climate change at 1.5°C. Science, 365, eaaw6974. https://doi.org/10.1126/science.aaw6974
- Horton, K. G., Sorte, F. A. L., Sheldon, D., Lin, T., Winner, K., Bernstein, G., et al. (2020). Phenology of nocturnal avian migration has shifted at the continental scale. *Nature Climate Chang*, 10(1), 63–68. https://doi.org/10.1038/s41558-019-0648-9
- Huang, J., Xie, Y., Guan, X., Li, D., & Ji, F. (2017). The dynamics of the warming hiatus over the Northern Hemisphere. *Climate Dynamics*, 48, 429–446. https://doi.org/10.1007/s00382-016-3085-8
- Kjellstrom, T. (2015). Impact of climate conditions on occupational health and related economic losses: a new feature of global and urban health in the context of climate change. Asia-Pacific Journal of Public Health, 28(2S), 28S–37S. https://doi.org/10.1177/1010539514568711
- Knudson, W. A. (2012). The economic impact of this spring's weather on the fruit and vegetable sectors (Working Paper 01-052012). East Lansing, MI: Michigan State University.
- Kral-O'Brien, K. C., O'Brien, P. L., & Harmon, J. P. (2019). Need for false spring research in the Northern Great Plains, USA. Agricultural & Environmental Letters, 4(1), 190025. https://doi.org/10.2134/ael2019.07.0025
- Menzel, A., & Fabian, P. (1999). Growing season extend in Europe. Nature, 397(6721), 659. https://doi.org/10.1038/17709
- Obradovich, N., Migliorini, R., Paulus, M. P., & Rahwan, I. (2018). Empirical evidence of mental health risks posed by climate change. Proceedings of the National Academy of Sciences, 115(43), 10953–10958. https://doi.org/10.7910/DVN/OVQY76
 - O'Neill, B. C., Tebaldi, C., Vuuren, D. P. V., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9(9), 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016
 - Orlov, A., Sillmann, J., & Vigo, I. (2020). Better seasonal forecasts for the renewable energy industry. *Nature Energy*, 5, 108–110. https://doi.org/10.1038/s41560-020-0561-5
 - Paniw, M., Maag, N., Cozzi1, G., Brock, T. C., & Ozgul, A. (2019). Life history responses of meerkats to seasonal changes in extreme environments. Science, 363, 631–635. https://doi.org/10.1126/science.aau5905
 - Park, B. J., Kim, Y. H., Min, S. K., & Lim, E. P. (2018). Anthropogenic and natural contributions to the lengthening of the summer season in the northern hemisphere. *Journal of Climate*, 31(17), 6803–6819. https://doi.org/10.1175/JCLI-D-17-0643.1

Acknowledgments

We acknowledge Tingjin Guan to draw Figure 5 and movie. Funding: Supported by the National Key R&D Program Grant 2018YFC1506903; the National Natural Science Foundation grants 41722502 and 41676021; Innovation Training Programs for Undergraduates, CAS. It is also the appropriate place to thank colleagues and other contributors. The authors thank the HadGHC-ND, CMIP5, and CMIP6 for providing accessible and excellent datasets that made this study possible.



- Peña-Ortiz, C., Barriopedro, D., & Garcia-Herrera, R. (2015). Multidecadal variability of the summer length in Europe. Journal of Climate, 28(13), 5375–5388. https://doi.org/10.1175/JCLI-D-14-00429.1
- Perkins-Kirkpatrick, S. E., & Lewis, S. C. (2020). Increasing trends in regional heatwaves. Nature Communications, 11, 3357. https://doi. org/10.1038/s41467-020-16970-7
- Power, S. B., & Delage, F. P. D. (2019). Setting and smashing extreme temperature records over the coming century. Nature Climate Chang, 9(7), 529–534. https://doi.org/10.1038/s41558-019-0498-5
- Qian, C., & Zhang, X. B. (2019). Changes in temperature seasonality in China: human influences and internal variability. Journal of Climate, 32(19), 6237–6249. https://doi.org/10.1175/JCLI-D-19-0081.1
- Reyes-Fox, M., Steltzer, H., Trlica, M. J., McMaster, G. S., Andales, A. A., LeCainD, R., & Morgan, J. A. (2014). Elevated CO2 further lengthens growing season under warming conditions. *Nature*, 510, 259–262. https://doi.org/10.1038/nature13207
- Riahi, K., Rao, S., Krey, V., Cho, C., Chirkov, V., Fischer, G., et al. (2011). RCP 8.5-A scenario of comparatively high greenhouse gas emissions. Climate Change, 109, 33–57. https://doi.org/10.1007/s10584-011-0149-y
- Ryan, S. J., Carlson, C. J., Mordecai, E. A., & Johnson, L. R. (2019). Global expansion and redistribution of Aedes-borne virus transmission risk with climate change. *PLoS Neglected Tropical Diseases*, 13(3), e0007213. https://doi.org/10.1371/journal.pntd.0007213
- Samplonius, J. M., & Both, C. (2019). Climate change may affect fatal competition between two bird species. *Current Biology*, 29, 327–331. https://doi.org/10.1016/j.cub.2018.11.063
- Santer, B. D., Po-Chedley, S., Zelinka, M. D., Cvijanovic, I., Bonfils, C., Durack, P. J., et al. (2018). Human influence on the seasonal cycle of tropospheric temperature. *Science*, 361, eaas8806. https://doi.org/10.1126/science.aas8806
- Schwartz, M. D., Ahas, R., & Aasa, A. (2006). Onset of spring starting earlier across the Northern Hemisphere. *Global Change Biology*, 12(2), 343–351. https://doi.org/10.1111/j.1365-2486.2005.01097.x
- Sheridan, S. C., & Lee, C. C. (2018). Temporal trends in absolute and relative extreme temperature events across North America. Journal of Geophysical Research: Atmospheres, 123, 11889–11898. https://doi.org/10.1029/2018JD029150
- Steltzer, H., & Post, E. (2009). Seasons and life cycles. Science, 324(5929), 886-887. https://doi.org/10.1126/science.1171542
- Stine, A. R., Huybers, P., & Fung, Y. I. (2009). Changes in the phase of the annual cycle of surface temperature. *Nature*, 457, 435–440. https://doi.org/10.1038/nature07675
- Thomson, A. M., Calvin, K. V., Smith, S. J., Kyle, G. P., Volke, A., Thomson, P. P., et al. (2011). RCP4.5: A pathway for stabilization of radiative forcing by 2100. *Climate Change*, 109, 77–94. https://doi.org/10.1007/s10584-011-0151-4
- Trenberth, K. E. (1983). What are the seasons? Bulletin of the American Meteorological Society, 64(11), 1276–1277. https://doi.org/10.1175/1520-0477(1983)0642.0.CO;2
- Westerling, A. L., Hidalgo, H. G., Cayan, D. R., & Swetnam, T. W. (2006). Warming and earlier spring increase western U.S. forest wildfire activity. Science, 313, 940–943. https://doi.org/10.1126/science.1128834
- Yan, Z., Xia, J., Qian, C., & Zhou, W. (2011). Changes in seasonal cycle and extremes in China during the period 1960–2008. Advances in Atmospheric Sciences, 28(2), 269–283. https://doi.org/10.1007/s00376-010-0006-3